Author's response to reviews

Title: A Randomised Controlled Trial of Ivabradine and Atorvastatin in Emergent Orthopaedic Lower Limb Surgery: A Mechanistic Study of Peri-Operative Myocardial Injury and its Prevention Using Computed Tomography Coronary Plaque Imaging and Novel Biomarkers of Cardiovascular Stress and Lipid Metabolism

Authors:

Nima Rudd (nima.rudd@nh.org.au)
Ivan Subiakto (ivansubiakto@gmail.com)
Vivek Mutha (vivekm@me.com)
Muhammad Asrar ul Haq (hissper@me.com)
William van Gaal (w.vangaal@nh.org.au)

Version: 2 Date: 5 May 2014

Author's response to reviews:

1. Key words moved below the Abstract.

2. Date the study was registered has been included with trial registration number at the end of Abstract.

3. Ethics statements have been included in the methods section:

"Northern Hospital Ethic Committee approved this study dated 6th July 2012. The study compies with the Helsinki Declaration. An informed consent will be obtained from the participating patients. Upon signing of the inform consent, patient’s data will be populated as per protocol. Currently the recruitment and patient enrollment is continuing."

4. Unavailability of funding has been acknowledged in the acknowledgement section.